Author's response to reviews

Title: Relationship between H.Pylori Infection and Clinicopathological Features and Prognosis of Gastric Cancer

Authors:

Hai-Bo Qiu Dr. (qiuhb@sysucc.org.cn)
Li-Yi Zhang Dr. (zhangliy@sysucc.org.cn)
Rajiv-Prasad Keshari Dr. (mayanksek@hotmail.com)
Guo-Qiang Wang Dr. (wqqjyh@163.com)
Zhi-Wei Zhou Prof. (zhouzhiw@mail.sysu.edu.cn)
Da-Zhi Xu (xdz1976@yahoo.com.cn)
Wei Wang (flwwj@163.com)
You-Qin Zhan (YQ_Zhan@21cn.com)
Wei Li (lwxsbs@medmail.com.cn)

Version: 4 Date: 1 December 2009

Author's response to reviews: see over
Relationship between H. Pylori Infection and Clinicopathological Features and Prognosis of Gastric Cancer

Qiu Hai-Bo\(^1,2\), Zhang Li-Yi\(^1,3\), Keshari Rajiv-Prasad\(^1,2\), Wang Guo-Qiang\(^1,2\),
Zhou Zhi-Wei\(^1,2\), Xu Da-Zhi\(^1,2\), Wang Wei\(^1,2\), Zhan You-Qin\(^1,2\), Li Wei\(^1,2\).

\(^1\)State Key Laboratory of Oncology in South China, Guangzhou, Guangdong, 510060, P. R. China
\(^2\)Department of Gastric and Pancreatic Surgery, Sun Yat-sen University Cancer Center, Guangzhou, Guangdong, 510060, P. R. China
\(^3\)Research Department, Sun Yat-sen University Cancer Center, Guangzhou, Guangdong, 510060, P. R. China

Qiu Hai-Bo and Zhang Li-Yi contributed equally to this work.
Correspondence to: Zhou Zhi-Wei
Tel: 86-20-87343626, Fax: 86-20-87343584, E-mail: zhouzhiw@mail.sysu.edu.cn

Grant Support:
Supported by Science and Technology Planning Project of Guangdong Province, China. (2008B030301103)
Dear Editor:
On behalf of my co-authors, I am submitting the enclosed material “Relationship between H.Pylori Infection and Clinicopathological Features and Prognosis of Gastric Cancer”. It is submitted to be considered for publication as an “original article” in the area of gastric cancer in your journal. This paper is new. Neither the entire paper nor any part of its content has been published or has been accepted elsewhere. It is not being submitted to any other journal.
We believe the paper may be of particular interest to the readers of your journal as its the basic findings of the paper and its significance. It known to us, gastric cancer patients with H.Pylori infection is very common in Asian country, However, At present, there is no convincing data for the relationship between H.Pylori infection status or copies and the clinicopathological features, besides, the prognosis differences in survival of patients who are positive for H.Pylori compared with those who are negative remains unknown. This study aimed to assess the relationship between H.Pylori and the clinicopathological features and prognosis of gastric cancer by quantitative detection of H.Pylori used the method of Real-Time PCR, and we found that the survival in patients with positive H.Pylori status had no significant difference from the patients with negative H.Pylori status. So we think that it may supply the evidence that the gastric cancer patients with H.Pylori infection have no need eradicate H.Pylori after surgery.
Thanks very much for your attention to our paper.
Sincerely yours,
Qiu Hai-B0
Address: Department of Gastric and Pancreatic Surgery surgery, Sun Yat-sen University Cancer Cencer, Guangzhou,Guangdong, 510060,P.R.China
Tel:86-20-87343123
Fax:86-20-87343584
Email: qiuhb@sysucc.org.cn